---
displayed_sidebar: homepage
---

import ParameterTypes from "@site/src/components/ParameterTypes"

# AbstractBatchJobStrategy

## constructor

### Parameters

<ParameterTypes parameters={[{"name":"__container__","type":"`any`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"__configModule__","type":"`Record<string, unknown>`","description":"","optional":true,"defaultValue":"","expandable":false,"children":[]},{"name":"__moduleDeclaration__","type":"`Record<string, unknown>`","description":"","optional":true,"defaultValue":"","expandable":false,"children":[]}]} expandUrl="https://docs.medusajs.com/development/entities/repositories#retrieving-a-list-of-records" sectionTitle="new AbstractBatchJobStrategy"/>

___

## Properties

<ParameterTypes parameters={[{"name":"_isBatchJobStrategy","type":"`boolean`","description":"","optional":false,"defaultValue":"true","expandable":false,"children":[]},{"name":"identifier","type":"`string`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"batchType","type":"`string`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"batchJobService_","type":"[BatchJobService](../../services/classes/services.BatchJobService.mdx)","description":"","optional":false,"defaultValue":"","expandable":false,"children":[{"name":"manager_","type":"`EntityManager`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"transactionManager_","type":"`undefined` \\| `EntityManager`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"__container__","type":"`any`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"Events","type":"`object`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"batchJobRepository_","type":"Repository&#60;[BatchJob](../../entities/classes/entities.BatchJob.mdx)&#62;","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"eventBus_","type":"[EventBusService](../../services/classes/services.EventBusService.mdx)","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"strategyResolver_","type":"[StrategyResolverService](../../services/classes/services.StrategyResolverService.mdx)","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"batchJobStatusMapToProps","type":"Map&#60;[BatchJobStatus](../enums/medusa.BatchJobStatus.mdx), object&#62;","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"__configModule__","type":"`Record<string, unknown>`","description":"","optional":true,"defaultValue":"","expandable":false,"children":[]},{"name":"__moduleDeclaration__","type":"`Record<string, unknown>`","description":"","optional":true,"defaultValue":"","expandable":false,"children":[]}]},{"name":"manager_","type":"`EntityManager`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"transactionManager_","type":"`undefined` \\| `EntityManager`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"__container__","type":"`any`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"__configModule__","type":"`Record<string, unknown>`","description":"","optional":true,"defaultValue":"","expandable":false,"children":[]},{"name":"__moduleDeclaration__","type":"`Record<string, unknown>`","description":"","optional":true,"defaultValue":"","expandable":false,"children":[]}]} expandUrl="https://docs.medusajs.com/development/entities/repositories#retrieving-a-list-of-records" sectionTitle="AbstractBatchJobStrategy"/>

___

## Accessors

### activeManager\_

#### Returns

<ParameterTypes parameters={[{"name":"EntityManager","type":"`EntityManager`","optional":false,"defaultValue":"","description":"","expandable":false,"children":[]}]} expandUrl="https://docs.medusajs.com/development/entities/repositories#retrieving-a-list-of-records" sectionTitle="activeManager_"/>

___

## Methods

### isBatchJobStrategy

#### Parameters

<ParameterTypes parameters={[{"name":"object","type":"`any`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]}]} expandUrl="https://docs.medusajs.com/development/entities/repositories#retrieving-a-list-of-records" sectionTitle="isBatchJobStrategy"/>

#### Returns

<ParameterTypes parameters={[{"name":"object","type":"object is IBatchJobStrategy","optional":false,"defaultValue":"","description":"","expandable":false,"children":[]}]} expandUrl="https://docs.medusajs.com/development/entities/repositories#retrieving-a-list-of-records" sectionTitle="isBatchJobStrategy"/>

### prepareBatchJobForProcessing

Method for preparing a batch job for processing

#### Parameters

<ParameterTypes parameters={[{"name":"batchJob","type":"[CreateBatchJobInput](../types/medusa.CreateBatchJobInput.mdx)","description":"","optional":false,"defaultValue":"","expandable":false,"children":[{"name":"type","type":"`string`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"context","type":"[BatchJob](../../entities/classes/entities.BatchJob.mdx)[`\"context\"`]","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"dry_run","type":"`boolean`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]}]},{"name":"req","type":"`Request`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]}]} expandUrl="https://docs.medusajs.com/development/entities/repositories#retrieving-a-list-of-records" sectionTitle="prepareBatchJobForProcessing"/>

#### Returns

<ParameterTypes parameters={[{"name":"Promise","type":"Promise&#60;[CreateBatchJobInput](../types/medusa.CreateBatchJobInput.mdx)&#62;","optional":false,"defaultValue":"","description":"Method for preparing a batch job for processing","expandable":false,"children":[{"name":"CreateBatchJobInput","type":"`object`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]}]}]} expandUrl="https://docs.medusajs.com/development/entities/repositories#retrieving-a-list-of-records" sectionTitle="prepareBatchJobForProcessing"/>

### preProcessBatchJob

Method for pre-processing a batch job

#### Parameters

<ParameterTypes parameters={[{"name":"batchJobId","type":"`string`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]}]} expandUrl="https://docs.medusajs.com/development/entities/repositories#retrieving-a-list-of-records" sectionTitle="preProcessBatchJob"/>

#### Returns

<ParameterTypes parameters={[{"name":"Promise","type":"Promise&#60;void&#62;","optional":false,"defaultValue":"","description":"Method for pre-processing a batch job","expandable":false,"children":[]}]} expandUrl="https://docs.medusajs.com/development/entities/repositories#retrieving-a-list-of-records" sectionTitle="preProcessBatchJob"/>

### processJob

Method does the actual processing of the job. Should report back on the progress of the operation.

#### Parameters

<ParameterTypes parameters={[{"name":"batchJobId","type":"`string`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]}]} expandUrl="https://docs.medusajs.com/development/entities/repositories#retrieving-a-list-of-records" sectionTitle="processJob"/>

#### Returns

<ParameterTypes parameters={[{"name":"Promise","type":"Promise&#60;void&#62;","optional":false,"defaultValue":"","description":"Method does the actual processing of the job. Should report back on the progress of the operation.","expandable":false,"children":[]}]} expandUrl="https://docs.medusajs.com/development/entities/repositories#retrieving-a-list-of-records" sectionTitle="processJob"/>

### buildTemplate

Builds and returns a template file that can be downloaded and filled in

#### Returns

<ParameterTypes parameters={[{"name":"Promise","type":"Promise&#60;string&#62;","optional":false,"defaultValue":"","description":"Builds and returns a template file that can be downloaded and filled in","expandable":false,"children":[{"name":"string","type":"`string`","optional":false,"defaultValue":"","description":"","expandable":false,"children":[]}]}]} expandUrl="https://docs.medusajs.com/development/entities/repositories#retrieving-a-list-of-records" sectionTitle="buildTemplate"/>

### shouldRetryOnProcessingError

#### Parameters

<ParameterTypes parameters={[{"name":"batchJob","type":"[BatchJob](../../entities/classes/entities.BatchJob.mdx)","description":"A Batch Job indicates an asynchronus task stored in the Medusa backend. Its status determines whether it has been executed or not.","optional":false,"defaultValue":"","expandable":false,"children":[{"name":"id","type":"`string`","description":"The unique identifier for the batch job.","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"created_at","type":"`Date`","description":"The date with timezone at which the resource was created.","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"updated_at","type":"`Date`","description":"The date with timezone at which the resource was last updated.","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"deleted_at","type":"`null` \\| `Date`","description":"The date with timezone at which the resource was deleted.","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"type","type":"`string`","description":"The type of batch job.","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"created_by","type":"`null` \\| `string`","description":"The unique identifier of the user that created the batch job.","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"created_by_user","type":"[User](../../entities/classes/entities.User.mdx)","description":"The details of the user that created the batch job.","optional":false,"defaultValue":"","expandable":true,"children":[]},{"name":"context","type":"`Record<string, unknown>`","description":"The context of the batch job, the type of the batch job determines what the context should contain.","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"result","type":"`object` & `Record<string, unknown>`","description":"The result of the batch job.","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"dry_run","type":"`boolean`","description":"Specify if the job must apply the modifications or not.","optional":false,"defaultValue":"false","expandable":false,"children":[]},{"name":"status","type":"[BatchJobStatus](../enums/medusa.BatchJobStatus.mdx)","description":"The status of the batch job.","optional":false,"defaultValue":"created","expandable":false,"children":[]},{"name":"pre_processed_at","type":"`Date`","description":"The date from which the job has been pre-processed.","optional":true,"defaultValue":"","expandable":false,"children":[]},{"name":"processing_at","type":"`Date`","description":"The date the job is processing at.","optional":true,"defaultValue":"","expandable":false,"children":[]},{"name":"confirmed_at","type":"`Date`","description":"The date when the confirmation has been done.","optional":true,"defaultValue":"","expandable":false,"children":[]},{"name":"completed_at","type":"`Date`","description":"The date of the completion.","optional":true,"defaultValue":"","expandable":false,"children":[]},{"name":"canceled_at","type":"`Date`","description":"The date of the concellation.","optional":true,"defaultValue":"","expandable":false,"children":[]},{"name":"failed_at","type":"`Date`","description":"The date when the job failed.","optional":true,"defaultValue":"","expandable":false,"children":[]}]},{"name":"err","type":"`unknown`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]}]} expandUrl="https://docs.medusajs.com/development/entities/repositories#retrieving-a-list-of-records" sectionTitle="shouldRetryOnProcessingError"/>

#### Returns

<ParameterTypes parameters={[{"name":"Promise","type":"Promise&#60;boolean&#62;","optional":false,"defaultValue":"","description":"","expandable":false,"children":[{"name":"boolean","type":"`boolean`","optional":false,"defaultValue":"","description":"","expandable":false,"children":[]}]}]} expandUrl="https://docs.medusajs.com/development/entities/repositories#retrieving-a-list-of-records" sectionTitle="shouldRetryOnProcessingError"/>

### handleProcessingError

#### Type Parameters

<ParameterTypes parameters={[{"name":"T","type":"`object`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]}]} expandUrl="https://docs.medusajs.com/development/entities/repositories#retrieving-a-list-of-records" sectionTitle="handleProcessingError"/>

#### Parameters

<ParameterTypes parameters={[{"name":"batchJobId","type":"`string`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"err","type":"`unknown`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"result","type":"T","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]}]} expandUrl="https://docs.medusajs.com/development/entities/repositories#retrieving-a-list-of-records" sectionTitle="handleProcessingError"/>

#### Returns

<ParameterTypes parameters={[{"name":"Promise","type":"Promise&#60;void&#62;","optional":false,"defaultValue":"","description":"","expandable":false,"children":[]}]} expandUrl="https://docs.medusajs.com/development/entities/repositories#retrieving-a-list-of-records" sectionTitle="handleProcessingError"/>

### withTransaction

#### Parameters

<ParameterTypes parameters={[{"name":"transactionManager","type":"`EntityManager`","description":"","optional":true,"defaultValue":"","expandable":false,"children":[]}]} expandUrl="https://docs.medusajs.com/development/entities/repositories#retrieving-a-list-of-records" sectionTitle="withTransaction"/>

#### Returns

<ParameterTypes parameters={[{"name":"this","type":"`this`","optional":false,"defaultValue":"","description":"","expandable":false,"children":[]}]} expandUrl="https://docs.medusajs.com/development/entities/repositories#retrieving-a-list-of-records" sectionTitle="withTransaction"/>

### shouldRetryTransaction\_

#### Parameters

<ParameterTypes parameters={[{"name":"err","type":"`Record<string, unknown>` \\| `object`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]}]} expandUrl="https://docs.medusajs.com/development/entities/repositories#retrieving-a-list-of-records" sectionTitle="shouldRetryTransaction_"/>

#### Returns

<ParameterTypes parameters={[{"name":"boolean","type":"`boolean`","optional":false,"defaultValue":"","description":"","expandable":false,"children":[]}]} expandUrl="https://docs.medusajs.com/development/entities/repositories#retrieving-a-list-of-records" sectionTitle="shouldRetryTransaction_"/>

### atomicPhase\_

Wraps some work within a transactional block. If the service already has
a transaction manager attached this will be reused, otherwise a new
transaction manager is created.

#### Type Parameters

<ParameterTypes parameters={[{"name":"TResult","type":"`object`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"TError","type":"`object`","description":"","optional":false,"defaultValue":"","expandable":false,"children":[]}]} expandUrl="https://docs.medusajs.com/development/entities/repositories#retrieving-a-list-of-records" sectionTitle="atomicPhase_"/>

#### Parameters

<ParameterTypes parameters={[{"name":"work","type":"(`transactionManager`: `EntityManager`) => Promise&#60;TResult&#62;","description":"the transactional work to be done","optional":false,"defaultValue":"","expandable":false,"children":[]},{"name":"isolationOrErrorHandler","type":"`IsolationLevel` \\| (`error`: TError) => Promise&#60;void \\| TResult&#62;","description":"the isolation level to be used for the work.","optional":true,"defaultValue":"","expandable":false,"children":[]},{"name":"maybeErrorHandlerOrDontFail","type":"(`error`: TError) => Promise&#60;void \\| TResult&#62;","description":"Potential error handler","optional":true,"defaultValue":"","expandable":false,"children":[]}]} expandUrl="https://docs.medusajs.com/development/entities/repositories#retrieving-a-list-of-records" sectionTitle="atomicPhase_"/>

#### Returns

<ParameterTypes parameters={[{"name":"Promise","type":"Promise&#60;TResult&#62;","optional":false,"defaultValue":"","description":"the result of the transactional work","expandable":false,"children":[{"name":"TResult","type":"TResult","optional":false,"defaultValue":"","description":"","expandable":false,"children":[]}]}]} expandUrl="https://docs.medusajs.com/development/entities/repositories#retrieving-a-list-of-records" sectionTitle="atomicPhase_"/>
